Who's Afraid of a Globalized World? Foreign Direct Investments, Local Knowledge and Allocation of Talents^{*}

Giovanni Pica[†] Università di Salerno and CSEF José V. Rodríguez Mora[‡] University of Southampton, UPF, CREA, CEPR, CES-ifo and IZA

June 4, 2007

Abstract

We study the distributional effects of globalization within a model of heterogeneous agents where both managerial talent and knowledge of the *local* economic environment are required in order to set up a firm in a given country. Therefore, agents willing to set up a firm in a foreign country need to incur a learning cost that depends on how different is the foreign entrepreneurial environments from the domestic one. In this context, we show that globalization fosters FDI and raises wages, output and productivity. Moreover, it benefits workers and highly talented multinational entrepreneurs, while harming low-ability domestic producers. The effects of openness follow from highly efficient foreign entrepreneurs driving inefficient local firms out of the market. We provide empirical evidence consistent with the implications of the model, showing a significant negative effect of the distance between nationwide regulations indexes on bilateral FDI flows.

JEL Classification: E61-F23-F41. **Keywords**: Multinational Firms, Heterogeneous Agents, Policy Harmonization

^{*}This is a revised version of a paper previously circulated under the title "Foreign Direct Investment, Allocation of Talents and Differences in Regulation" (CEPR WP 5318) prepared for the *Second Annual Conference of the Euro-Latin Study Network on Integration and Trade* (ELSNIT), Florence, October 29-30, 2004. We are grateful to Chiara Fumagalli, Tony Venables, and Thierry Verdier for useful comments. We also thank the audience of seminars at the European University Institute, University of Southampton, University of Prague, Università Bicocca of Milan. Part of this paper was written while the first author was visiting the European University Institute whose hospitality is gratefully acknowledged. Any remaining errors and omissions are our responsibility.

[†]E-mail: gpica@unisa.it

[‡]E-mail: sevimora@gmail.com

1 Introduction

Foreign Direct Investments (FDI) are one of the predominant and most debated features of globalization. FDI grew dramatically in the last 15 years of the 20th century, far outpacing the growth of trade and income: whereas world-wide real GDP increased at a rate of 2.5 percent per year between 1985 and 1999 and world-wide exports by 5.6 percent, world-wide real inflows of FDI increased by 17.7 percent. Another salient feature of FDI is that they take place mostly between developed countries, i.e. between countries that are similar in terms of natural endowments and relative supply of inputs. For the period 1970-2000, Barba Navaretti and Venables (2004) report that more than 90% of outward flows of FDI originates from advanced countries. Over the same period, the share of the world FDI inflows directed to developed countries ranges between 58 and 78 percent.¹ Moreover, a growing body of empirical evidence is pointing to the existence of a positive effect of FDI inflows on wages and productivity. Baldwin, Braconier and Forslid (1999) show that FDI positively affect wages using industry-level data for seven OECD countries. Keller and Yeaple (2003) provide firm-level evidence from the US showing that FDI spillovers account for about 14% of productivity growth in U.S. firms between 1987 and 1996. Javorcik (2004) provides similar evidence for Lithuania.²

In this paper, we build a very simple general equilibrium model that provides an encompassing explanation of these facts and, at the same time, allows to study the distributional effects of globalization through FDI. In fact, even though there is a general consensus that the process of globalization is bound to be in the long-run welfare improving, it may still produce winners and losers if appropriate transfer schemes are not available. This issue, despite its obvious relevance, has been so far hardly analyzed in the literature.

To this aim, we consider a world of heterogeneous agents where both managerial talent and knowledge of the local economic environment are required in order to set up a firm and earn positive profits. The main trade-off that arises in the model depends on how individuals with different abilities are allocated to different types of jobs available in the economy.

To be more specific, a first key feature of the model is that agents with different levels of managerial ability are allowed to choose their occupation, i.e. whether to become entrepreneurs or workers. Those who become entrepreneurs may, in turn, engage in FDI, i.e. set up a firm abroad. However, in order to become a successful entrepreneur in a given country, managerial ability is not sufficient: some *knowledge* of the local economic environment is also required. A second key feature of the model is that domestic agents are assumed to know the characteristics of the domestic economic environment (e.g. domestic consumers' tastes) while they ignore the characteristics of the foreign economic environment and have to *learn* them if they want to profitably set up a firm abroad. Thus, both managerial ability *and* nationality contribute to determine career choices. The idea is that a certain level of managerial talent, though allowing

¹Surprisingly, economic theory has been geared mainly towards explaining flows of capital (and goods) among countries that are different. This is clearly the case for Ricardian and Heckscher-Ohlinean models while "new-trade" models \dot{a} la Krugman explain intraindustry trade and investment flows between economies that are *ex-ante* similar and *ex-post* different: increasing returns to scale favor specialization and concentration of production (and, thus, trade). Locations are *ex-post* different because, once specialization has taken place as the consequence of increasing returns, different varieties are produced in each country.

²See Lipsey (2002) for a review of the micro evidence on the home and host country effects of FDI.

agents to profitably produce within the economic domestic environment, may be of little help when setting up a firm abroad. The more so, the more different the foreign and the domestic economic environments. This *distance* between entrepreneurial environments is the only explicit barrier to capital movements that matters in the model. It may be overcome only at the cost of learning how the foreign environment works. Of course, in equilibrium, only the most talented entrepreneurs have incentives to pay the learning cost and produce abroad. Thus, the model endogenously determines the allocation of talents between (domestic and international) entrepreneurial activity and salaried work. It follows that the pattern of FDI flows and the levels of Total Factor Productivity (TFP), GDP and wages depend on how efficiently talents are allocated. Talent allocation, in turn, depends on how hard it is to learn how to deal with the foreign entrepreneurial environment. A lower distance between entrepreneurial environments reduces the learning cost and raises the inflow of foreign-owned firms into the local market. This increases local wages and makes the entrepreneurial activity less profitable, driving a fraction of low-ability domestic entrepreneurs out of the market. This general equilibrium effect improves the allocation of talents and increases both TFP and GDP. On the contrary, a larger distance between entrepreneurial environments protect low-ability entrepreneurs from foreign competitors and reduce output, wages and TFP.

Summing up, this paper makes the following three points.

- 1. Globalization, by reducing the distance between entrepreneurial environments (for example, by making consumers' taste more homogeneous throughout the world), raises aggregate GDP and TFP, by inducing larger FDI flows. The underlying mechanism works through the improved allocation of talents in the economy. Thus, globalization fosters efficiency.
- 2. Globalization has large distributional effects. The individuals with the lowest level of entrepreneurial talent (who choose to become workers independently of the level of globalization) and the individuals with the highest level of talent (who choose to become multinational entrepreneurs if the degree of globalization is high enough) are better off in a fully globalized universe, i.e. an universe where the learning cost is zero and therefore GDP, TFP and wages are largest.

Differently, the individuals with an "intermediate" level of talent (i.e. local entrepreneurs who never choose to engage in FDI and may be even driven out of the entrepreneurial activity if the degree of globalization is high enough) prefer to live in a non-globalized universe. The reason is that, in a globalized world, they pay the cost of tougher competition without enjoying the benefits of accessing to wider markets. Only in a non-globalized world they survive as entrepreneurs as they are sheltered from foreign competitors.

3. Finally, in our model, a lower cross-country distance between entrepreneurial environments raises the volume of bilateral FDI flows. We test this prediction against the data. We proxy the entrepreneurial environment using OECD and the World Bank indexes that measure the level of Product Market Regulation. We match these data with data on bilateral FDI flows drawn from the *OECD International Direct Investment Statistics*.

Our empirical specification lies in the tradition of gravity models with an additional ex-

planatory variable: the *absolute value* of the difference between the source and host country indexes of regulation. We find that, controlling for the *level* of regulation in both countries, countries fixed effects, time effects, countries GDP and a set of geographical variables, the coefficient of the variable capturing regulation proximity is negative and in most cases significant. We interpret this finding as evidence consistent with the hypothesis that the distance between nationwide regulations contributes to shape the size of bilateral FDI flows. Thus, we find our empirical results suggestive that similarity in entrepreneurial environment fosters FDI.

Even though these points are strictly interrelated, for the sake of clarity it is convenient to discuss them (and relate them to the literature) separately. We do so in section 2. In particular, subsection 2.1 discusses the literature on the determinants of FDI, subsection 2.2 the literature on allocation of talents, and subsection 2.3 the literature on the distributional effects of globalization. The rest of the paper is organized as follows. Section 3 presents the data and shows that cross-country differences in regulation affect FDI flows. Section 4 describes the model economy. Section 5 solves for the closed economy benchmark and section 6 analyzes the framework where entrepreneurs are allowed to set up firms abroad. Finally, section 7 concludes.

2 Related Literature

2.1 Cross-country differences and FDI

An extensive (theoretical and empirical) literature has studied the factors that drive FDI flows.³ The key findings may be quickly summarized as follows.

First, a host of institutional, technological and market factors affect firms' decision to set up production facilities in a foreign market.⁴ Obvious examples are trade barriers, (firm- vs. plantlevel) economies of scale, and market size.⁵ Second, larger cross-country factor cost differentials, generated either by differences in productivity or in the relative supply of inputs, are typically associated to larger FDI flows.⁶ Third, and more importantly for this paper, *larger* cross-country differences along some well identified dimensions, negatively affect FDI flows. Variables like "smaller physical distance", "sharing a common language" or "sharing a border" significantly contribute to determine the size of bilateral FDI flows in gravity-like empirical models (see Ekholm (1998) and Shatz (2003)). Moreover, Markusen and Maskus (2002) show that the

 $^{^{3}}$ See among others Horst (1972), Deardorff, (1998), Ekholm (1998), Lipsey (2001), Razin et al. (2003), Shatz (2003).

 $^{{}^{4}}$ See Barba Navaretti and Venables (Chap 2, 2004) for an in-depth discussion of the benefits and costs to the firm of FDI.

⁵First, trade barriers encourage *horizontal* FDI, i.e. FDI aimed at serving a local market, while they discourage *vertical* FDI, i.e. FDI aimed at reducing production costs by relocating (part of) the production activities abroad. Second, multinational firms are likely to be characterized by intangible firm-specific assets from which firm-level (as opposed to plant-level) scale economies originate. Finally, most FDI are directed toward large markets: as investing in a given country implies large fixed costs, firms are willing to afford it if perspective sales are sufficiently large (Brainard (1997)).

⁶Vertical FDI, that involve the fragmentation of the production process and the relocation of the most labor intensive technologies to relatively low wage countries, typically exploit factor prices differences. Using US data, Brainard (1993) and Markusen and Maskus (2001) get some support for the relevance of factor market considerations as determinants of FDI.

similarity between host and home factor endowments is one of the main factors driving the location of foreign subsidiaries (together with the size of the host market).

We focus our attention on the third set of determinants of FDI and (1) provide an explanation of why smaller cross-country differences (along some *non obvious* dimensions) foster bilateral FDI flows and (2) produce new empirical evidence in support of this finding.⁷

In our model countries will not differ in terms of factor endowments. Cross-country differences will rather lie in a factor that we call "entrepreneurial environment". The following statement from Unilever's website helps understanding what we mean by "entrepreneurial environment":

"Many of our brands have international appeal, while others are leaders in local markets. It is our *keen understanding of cultures and markets* that allows us to anticipate consumers' needs and to provide them with what they need, when they need it." (Unilever, emphasis added)⁸

Broadly speaking, one may think of the entrepreneurial environment as representing the complex set of circumstances, generally different across countries, entrepreneurs need to deal with: identification of consumers' tastes, communication with costumers, relationship with the bureaucracy, comprehension of the legal environment, purchase of inputs, relationship with other firms, setup of the production process (hiring and firing procedures, salary structure, technology choices,...).⁹

Given this, our reasoning is simple: in each country only some agents become entrepreneurs. These are the ones that have a combination of managerial ability and knowledge of the local environment that allows them to profitably produce and sell in the domestic country. However, at least to some extent, knowledge of the local environment also allows to infer the characteristics of the foreign economic environment. The more so the more similar the two countries. It follows that smaller cross-country differences between entrepreneurial environments make it easier for domestic entrepreneurs to set up firms abroad.

While we describe this mechanism in a formal general equilibrium model (sections 5 and 6), our contribution, in this respect, is also empirical. We regress bilateral FDI flows on (a set of controls and) variables that proxy the distance between "entrepreneurial environments". The variables that we use are indexes that measure the cross-country distance between the levels of Product Market Regulation. Usually in empirical works, proximity in "regulations" is not taken into account as one of the possible determinants of FDI. However, our model shows that it should matter. The reason is that national regulations contribute to shape the economic environment, because they typically prescribe to follow particular procedures (e.g. business start-up procedures, administrative rules, safety and health regulations, food regulations). The

⁷The reason why we do not focus on the first two sets of determinants of FDI is twofold. First, the channels through which they shape *Ricardian* (and *"Hecksher-Ohlinean"*) FDI flows are well understood. Second, the overwhelming proportion of FDI is horizontal rather than vertical and this implies that typically FDI do not flow in order to exploit factor price differences.

⁸Cite taken from Barba Navaretti and Venables (2004).

⁹Thus, the entrepreneurial environment is shaped by demand- and supply-side factors. Demand factors are related to the environment of the downstream market where final transactions take place, while supply factors are related to the environment where *production* take place.

more the institutional settings (or any kind of law that imposes to comply with some procedures) are different, the more costly the adaptation process to the new environment and the smaller the incentives to actually run businesses abroad. Thus, rough as it may be, this measure captures (at least part of) the difference between entrepreneurial environments and has the advantage of being easily observable. Interestingly, we find that the coefficient of the variable capturing regulation proximity is negative and typically significant. Thus, similarity in the levels of nationwide regulations does seem to contribute to raise the size of bilateral FDI flows.

We now discuss the role of agents' heterogeneity and how our model relates to the literature on the allocation of talents.

2.2 Heterogeneity and Allocation of Talents

Our paper is strictly related to the seminal paper of Melitz (2003), who develops a dynamic industry model with heterogeneous firms to analyze the intra-industry effects of international trade. As in our paper, also in Melitz (2003) only the most efficient firms export and exposure to trade forces the least productive firms to exit the market via higher real wages. The key difference is that in Melitz (2003) heterogeneity is cast at the firm level: consumers are homogeneous and there is no endogenous sorting of agents into jobs. In our paper, firms' heterogeneity stems from the heterogeneity (in managerial talent) of the agents, who are allowed to make career choices. These features of our model allow to stress the role of the (endogenous) mechanism by which exposure to foreign competition improves the allocation of talents and, most importantly, to discuss the distributional implications of globalization.

In order to make the differences between Melitz (2003) and the present paper clear, let us stress (again) the link (absent in Melitz, 2003) between globalization and allocation of talents *via* FDI. As already explained, in our model agents have both different levels of managerial ability and different nationalities. Nationality matters because, given a certain level of managerial ability, it gives an advantage *vis-à-vis* foreign entrepreneurs thanks to a better knowledge of the local economic environment. Once individuals are allowed to make career choices, in each country the pool of entrepreneurs will consist of individuals with *relatively* high entrepreneurial talent. Within the pool of entrepreneurs, only the most talented will set up firms (not only at home but also) abroad. The issue is *how much* talent is required in order to become a local/multinational entrepreneur. It obviously depends on how costly it is to be an entrepreneur and, in particular, on how costly it is to hire labor.

We make a very simple point. Globalization, by reducing the cost of acquiring information about the foreign country, spurs FDI flows, increases the demand for labor and, consequently, raises wages. The increase in wages has two effects. (1) It reduces firms' profits, making the entrepreneurial activity less attractive *per se.* (2) It increases the opportunity costs of being an entrepreneur, making the alternative choice (be a worker) more attractive. The obvious consequence is that larger FDI flows, by pushing up the cost of labor, drive the least talented domestic producers out of the entrepreneurial activity. Notice that this happens even in the absence of the standard pro-competitive effect of FDI that works through lower prices;¹⁰ only

¹⁰The competition effect is present in almost all the standard IO-based FDI models, since Horstmann and

the general equilibrium effect, via higher labor demand and higher wages, is at work.

Allocation of talents and FDI flows are therefore strictly related. As only the most efficient entrepreneurs can afford the cost of learning how the foreign economic environment works and engage in FDI, larger FDI flows directly improve the domestic allocation of talents and raise aggregate efficiency. Moreover, the effect on efficiency is further reinforced by the fact that larger FDI flows drive bad domestic entrepreneurs out of the market.

A further point that differentiates this paper from Melitz (2003) is the analysis of the distributional implications of globalization, to which we now turn.

2.3 The Effects of Globalization: Winners and Losers

To the best of our knowledge, this is the first paper that uncovers the distributional effects of globalization. For the sake of clarity, let us consider two polar cases. The case of a *Globalized Universe*, where the entrepreneurial environment is identical across locations and therefore nationality is irrelevant; and the case of a *National Universe* where entrepreneurial environments differ and nationality is a relevant characteristic.

(1) Globalized Universe. In this world only talent matters: learning costs are zero. The most talented individuals become entrepreneurs and there exists a critical level of talent that makes the marginal individual indifferent between being an entrepreneur or a worker. An individual whose entrepreneurial talent lies just below that critical level would choose to be an entrepreneur only if wages were lower, both because profits would be higher and because the worker option would be less attractive.

(2) National Universe. In this world the entrepreneurial environments are different across countries and learning how the foreign environment works is costly. For the sake of clarity, let us consider the case where the cost of learning how the foreign environment works is infinitely large so that FDI are *de facto* ruled out. In this world only local firms demand labor in the local labor market. Consequently, wages are lower compared to the *Globalized Universe*.

Consider now three individuals. In decreasing order of managerial ability:¹¹ Ms Capitalistson, Ms Petitbourgeoison and Ms Proletariatson.

The first one (Ms. Capitalistson) has a large degree of entrepreneurial talent. In the *Globalized Universe* she invests both at home and abroad while in the *National Universe* she is a domestic entrepreneur. On the one hand, she likes the *National Universe* because wages are lower and this implies larger domestic profits. On the other hand, she also likes the *Globalized Universe* because of the larger investment possibilities. It turns out that if her talent is large

Markusen (1992). In our model product market competition does *not* increase in the domestic country as a consequence of foreign competition. We rule this effect out by assuming monopolistic competition and Dixit-Stiglitz preferences.

¹¹By managerial ability we mean any individual characteristic that helps being a successful entrepreneur. Talent is one, though not the only one. The ability of raising funds in a world with capital market imperfections is certainly another; in that case having wealthy parents would be akin to being more talented. In this paper we will consider managerial ability as exogenous and independent of the economic environment (the degree of competition, etc.). To see a model where this is not the case (i.e., where the contributions of talent and family background depend on equilibrium outcomes) see Hassler and Rodríguez Mora (2000). In any case, there is nothing offensive about having less managerial talent in our model.

enough she prefers the Globalized Universe.

The second one (Ms. Petitbourgeoison) has a lower degree of entrepreneurial ability. Her talent level is such that, in the *Globalized Universe*, she (slightly) prefers to be a worker. Therefore, in the *National Universe* where wages are lower, she chooses to become entrepreneur. Thus, globalization expels her from the entrepreneurial activity and makes her strictly worse off.¹² The intuition is simple: from an entrepreneur's point of view, globalization is beneficial insofar it allows to gain access to larger markets. Low-ability entrepreneurs lose from globalization because tougher competition drives them out of the market and prevents them from reaping the benefits of accessing to larger markets.

Finally, Ms. Proletariatson has even lower entrepreneurial abilities. So low that she chooses to be a worker independently of the world (*Globalized* or *National*) where she happens to live. It follows that she prefers to live in the *Globalized Universe*, where wages are higher.

Thus, in a national world differences in entrepreneurial environments shelter inefficient firms. Even if there is no direct pro-competition effect of FDI, the general equilibrium effect on the labor market is sufficient to expel mediocre entrepreneurs as differences in the environments become smaller. Our model predicts that the middle-ability class (i.e. local entrepreneurs that would not invest abroad anyway) always supports localisms and regional fragmentation while the end-tails of the ability distribution (i.e. workers and producers willing to engage in FDI) would rather live in a globalized universe.¹³

3 Cross Countries Differences and FDI

The argument presented above (formally presented in the model outlined in section 4) implies that a lower cross-country distance between economic environments leads to higher bilateral FDI flows, because it reduces the cost of learning how to deal with the foreign entrepreneurial environment. This sections provides empirical evidence in this direction. Specifically, it shows a negative relation between bilateral FDI flows and variables that proxy the cross-country distance between entrepreneurial environments.

From an empirical point of view, a major difficulty is that the "distance between entrepreneurial environments" is difficult to measure. A good proxy, however, may be represented by the different levels (and types) of regulations implemented in different countries. The idea is that regulation, along its several dimensions, is one of the key determinants, although of course not the only one, of the entrepreneurial environment. Thus, our working hypothesis will be that cross-country differences in the extent of regulation translate into differences in entrepreneurial environments. To conduct the analysis, we exploit two datasets, collected by the OECD and the World Bank, providing country-level regulations indexes and data on bilateral FDI flows.

Panel (a) in figure 1 displays on the horizontal axis the difference between country i and country j index of *Barriers to International Trade and Investment* in the late 90's, as measured

 $^{^{12}}$ To see that Ms. Petitbourgeoison is strictly worse off under globalization notice that she is almost indifferent between career choices in the *Globalized Universe* and becomes entrepreneur in the *National Universe*. Lower wages imply that entrepreneurial profits are larger in a *National world*, so she must be better off in such a world.

 $^{^{13}}$ Section 6.4 elaborates more on this point.



Figure 1: Plots of the regulatory distances between countries i and j against the FDI flows from country j to country i (1980-1997).

by the OECD, and on the vertical axis the flow of FDI from country j to country i (from 1980 to 1997). The graph shows that bulk of FDI flows lies in the area where the difference between regulations is close to zero. The smaller the difference between regulations, the larger and the more frequent the bilateral flows of FDI. Panel (b) in figure 1 uses a different measure of regulation, *Barriers to Entrepreneurship*, on the horizontal axis. On the vertical axis, as before, the flow of FDI from country j to country i. Again, a smaller difference between regulations tends to be associated with larger bilateral FDI flows. Finally, Panel (c) in figure 1 considers as a third measure of regulation the extent of *state control over business enterprises*. Also this graph shows the "triangular" pattern as the previous ones.

These graphs suggest that FDI do *not* flow from more regulated countries (where one would tend to think that the rewards from capital are low^{14}) to less regulated economies where one would think that the rewards from capital are high. If this was the case, we would observe much larger and frequent FDI flows when the difference between the regulation indexes is negative. Rather, it seems that smaller differences in regulations tend to be associated with larger bilateral FDI flows.¹⁵

Before moving to the regression-based evidence we turn to the description of the data.

3.1 The Data

The data on FDI are drawn from the *OECD International Direct Investment Statistics* that provides yearly statistics for OECD countries on international direct investment flows (inflows, outflows) by geographical distribution, i.e. to and from partner countries and regions from 1980 to 1997. Data are provided in national currency and have been converted to US dollars using yearly average exchange rates.

As control variables we use GDP, from the OECD Main Economic Indicators, population

¹⁴Unless, of course, regulated countries are poor and have a high marginal productivity of capital. In this case we should expect these countries to enjoy net FDI inflows. Our point is that, *in addition to this effect*, the flows seem to depend negatively on the regulatory *distance*.

¹⁵One alternative explanation may be that the negative relation between differences in regulations and FDI flows is due to the fact that FDI take place mostly among (rich) non regulated countries that have *similar* regulation levels. If this was the case, what would actually foster bilateral FDI is not the regulation proximity between countries but, rather, their low regulation levels. The empirical analysis that follows controls for the *level* of regulation and GDP in each country and therefore allows to reject this hypothesis.

from the Penn World Tables, and a set of geographic variables that includes: latitude and longitude of the source and host country; an adjacency dummy (i.e. if countries share common land borders); a linguistic tie dummy (i.e. if countries share a common language); distance between (the main cities of the) countries; European Union, North America and Asian dummies; a NAFTA dummy. The geographical variables are drawn from Frankel, Stein and Wei (1995) and Frankel and Wei (1998).

Finally, we exploit variables capturing the level of different types of regulation implemented in different countries. We use two sets of such variables, one from the OECD (Nicoletti et al. (2000)) and one from the World Bank (available at http://www.doingbusiness.org/).

The OECD dataset consists of indexes measuring the extent of Product Market Regulation in a number of OECD countries during the 90's. It provides both an overall index of Product Market Regulation and a set of sub-indexes measuring the extent of regulation along particular dimensions, namely *Barriers to international trade and investment*, *Barriers to entrepreneurship*, *State control over business enterprises*, *Administrative regulations*, *Economic regulation*. Of particular interest for our purposes are the indexes capturing mostly administrative burdens and red tape costs (*Administrative regulations* and *Barriers to entrepreneurship*), i.e. all those bureaucratic procedures whose *knowledge* is an essential prerequisite in order to be able to set up a firm in a (foreign) country. In what follows we will be exploiting the whole set of sub-indexes. Though some of them are clearly highly correlated, others, like *Administrative regulations* and *Barriers to international trade and investment* display very low correlation (see Table 2 in the appendix).

The World Bank, on the other side, provides a comprehensive database, called *Doing Business*, collecting information on business regulations and their enforcement, especially on smalland medium-size domestic firms, for 145 countries. The dataset we exploit refers to January 2004. The available indicators cover seven major areas, namely Starting a Business, Hiring and Firing, Registering Property, Getting Credit, Protecting Investors, Enforcing Contracts, and Closing a Business. For each of them different indexes are provided. Some indicators (like *Number of procedures to register a business* or *Index of employment law rigidity*) aim at measuring the effect of *actual* regulation on businesses, while others (such as *Time and cost to register a business, enforce a contract, or go through bankruptcy*) are measures of regulatory *outcomes*. Table 3, in the appendix, displays the full set of available variables.

Of course, these measures of regulation are far from perfect. Ideally, one would like to have time-varying information on whether regulations are *qualitatively* different among countries rather than just *quantitatively* different. The problem may be that two countries where the number of procedures to start up a business is the same may require very different tasks to be complied with. However, since quantitative differences in the amount of regulation plausibly come together with qualitative differences, these data allow to capture, at least partially, qualitative differences as well.

3.2 Empirical Results

In order to check whether the patterns highlighted by the graphical analysis are robust to a more thorough empirical analysis, we run a battery of regressions in the tradition of gravity models, adding additional explanatory variables like the regulation *level* and the *absolute value* of the difference between the source and host country indexes of regulation.

We estimate the following model:

$$F_{ijt} = \alpha_i + \tau_t + X_{ijt}\beta + \gamma |reg_i - reg_j| + \varepsilon_{ijt}$$
(1)

where F_{ijt} is the FDI flow from country j (the source) to country i (the host) at time t, as a share of the GDP of the host country; α_i is a (host) country fixed effect; τ_t is a year effect; the vector X_{ijt} includes variables, such as the source and host countries GDP (in US dollars); the source and host countries population; the latitude and longitude of the source country;¹⁶ the distance between the main cities of the two countries. Dummy variables are included to control for country i and j sharing the same language; sharing common land borders; both belonging to the European Union; being both located in North America; being both located in Asia; both belonging to the NAFTA. These geographical variables are meant to capture the proximity-concentration trade-off (Brainard (1997)).

Finally, we control for the *level* of regulation in both the host and source country by exploiting a time-varying index of Employment Protection Legislation measured (Nicoletti et al. (2000)) in 1990 and 1998. This is the only available measure of regulation that varies over time. It allows to control for the level of regulation also in the host country where non time-varying characteristics are otherwise captured by the fixed effect.¹⁷

Controlling for the level of regulation, the coefficient γ captures exclusively the effect of *regulation proximity*, as measured by the absolute value of the difference between regulation indexes, on FDI bilateral flows.

We perform our analysis on a set of 22 countries, for which we have data on both regulation indexes and FDI flows, listed in Table 1 (see appendix). Tables 4, 5, 6 and 7 report the results of the estimation.

Table 4: results using the OECD regulation data

Table 4 presents the results from the estimation of equation (1) obtained exploiting the OECD regulation variables. Columns A–H report the results of eight alternative specifications differing in the measure of regulation proximity included on the right hand side. In column A we use the overall index of Product Market Regulation and, from column B to column H, we exploit the sub-indexes that focus on particular dimensions of product market regulation. In each specification, the regulation *levels* of the source and host country are controlled for by the time-varying regulation measure of EPL described above.

The first two rows of table 4 report the coefficients of the regulation levels that have, as

¹⁶The latitude and longitude of the host country, as well as any other non time-varying characteristic of the host country, are captured by the fixed country effect α_i .

¹⁷Results are unchanged if one controls for the regulation level exploiting other measures of regulation.

expected, negative signs.¹⁸ The remaining rows of table 4 show the coefficients of the different indexes that measure the distance between regulations. The coefficients of *all* variables measuring the regulatory distance across countries are negative and significant at the conventional significance levels.

Tables 5, 6 and 7: results using the World Bank regulation data

These tables report the results from the estimation of equation (1) using the World Bank *Doing Business 2004* dataset. Overall, the majority of the World Bank indexes of regulation proximity are both negative and significant. Some, though all negative (except one), are not significant. In particular, Table 5 shows that the distance in regulations that measure the difficulty of *Starting a Business* and the difficulty of *Hiring and Firing* are all negative and significant (except the variable *Minimum capital to start a business (% of income per capita)*).

Table 6 shows that higher similarity in regulations concerning *Property Registration* also has a positive effect of FDI inflows as both the *Number of procedures to register a property* and the *Number of days to register a property* enter negatively and significantly. Differently, the results on the effect of differences in the *Credit* system are mixed as the *Index of legal rights of borrowers and lenders* and *Coverage of private registry* are negative and significant while the coefficients of the other variables, though negative, are not significant. This may reflect the fact the multinationals have easier access to credit to and may also rely on internal capital markets.

Finally, Table 7 suggests that, while differences in the index of *Investor Protection* is not significant, a larger distance in the procedures related to *Contract Enforcement* reduces the inflows of FDI. Finally, larger differences in the *Bankruptcy* procedures do not matter for FDI.¹⁹

Economic significance

As our regulation variables are, in many cases, indexes with no natural scale, the magnitude of the coefficients is not *per se* informative of the potential impact of regulation proximity on FDI flows. Therefore, we have calculated the so-called "beta" coefficients, reported in table 8 along with the sample means and standard deviations. A beta coefficient is given by the product of the estimated coefficient and the standard deviation of its corresponding independent variable, divided by the standard deviation of the dependent variable. It converts the regression coefficients into units of sample standard deviations and is equivalent to a regression where all variables are previously divided by their standard deviations.²⁰

The magnitude of the beta coefficients reported in table 8 suggests that regulation proximity has a non negligible impact on FDI flows. For instance, a one standard deviation decline in the distance between *Barriers to Entrepreneurship* raises the ratio of FDI to GDP of as much as 7.81% of its standard deviation. This number is not small as, in the same regression, the beta coefficient of the linguistic tie dummy (sharing the same language) is a very similar 0.08. The beta coefficients of the remaining independent variables in each of the different specifications

¹⁸Only the coefficient of the source country regulation level is statistically significant, because of the presence of the host country fixed effect and the little time variation of the regulation index.

¹⁹Also in tables 5, 6 and 7 the regulation *levels* enter with a negative sign for both countries, but only the coefficients of the source country are statistically significant.

²⁰Wooldridge (2003, Section 6.1).

are not reported for brevity and are available upon request. However, the broad picture that emerges suggests that the magnitude of the effect of the regulation proximity variables is, in general, comparable to that of the other proximity variables usually included in gravity-like models.

Therefore, the overall evidence suggests that, along some dimensions, differences in regulations may matter i.e., controlling for the level of regulation, a smaller distance between national regulations fosters bilateral FDI flows. In particular, we find that the distance between regulations concerning *Product Markets, Labor Markets* (with some emphasis to be placed on firing restrictions), and *Contract Enforcement* plays a prominent role in shaping bilateral FDI flows. Notice that these regulations have to do with the way entrepreneurs have to set up firms. For example, product market regulations typically imply the existence of particular start up procedures one needs to comply with in order to start a business. Labor market and contract enforcement regulations may forbid (or allow only for) certain types of contractual arrangements to be stipulated with workers and/or suppliers. Thus, entrepreneurs need to *know* these prescriptions in order to set up a firm.

The next section presents a simple general equilibrium model, consistent with the above empirical evidence, that allows to study the distributional effects of globalization.

4 The Model

4.1 Demand and Production

There are two political entities (countries). In each of them agents have Dixit-Stiglitz preferences on the mass of products sold in their country. The demand for good j is:

$$x_j = Y p_j^{-\theta}$$

where Y stands for aggregate demand in the country, θ is the constant demand elasticity and p_j is the price of the good. We normalize the price of the "aggregate" good in each country to 1. All goods are consumed in the country where they are produced.²¹ Entrepreneurs (who can be either national or foreigner) set up firms and face a monopolistic environment. All firms produce with constant returns to scale using only labor. They maximize:

$$\max_{p_j} \left(p_j x_j - a w x_j \right) = Y p_j^{-\theta} \left(p_j - a w \right)$$

a being unit labor requirements. Thus, gross profits (gross of fixed costs, as explained below) and productive labor demand (*productive* as opposed to *total* labor demand, as again explained

 $^{^{21}}$ In appendix C we relax this assumption and assume that a tradable good is produced by perfectly competitive firms in the context of a two-sector model.

below) of a single firm are respectively:

$$\pi = Y\left(\frac{\theta}{\theta-1}\right)^{-\theta} (wa)^{1-\theta} \frac{1}{\theta-1}$$
(2)

$$z = \left(\frac{\theta}{\theta - 1}\right)^{-\theta} a^{1 - \theta} w^{-\theta} Y \tag{3}$$

In the rest of the paper we assume without loss of generality that $\theta = 2$ and $a = \frac{1}{4}$. Gross profits and productive labor demand, then, simplify to:

$$\pi = \frac{Y}{w} \tag{4}$$

$$z = \frac{Y}{w^2} \tag{5}$$

4.2 Entrepreneurs and net profits

Agents are heterogeneous and differ in their ability to run businesses. Each agent faces a career choice. She has to decide whether to become a worker or an entrepreneur. Agents choosing to become entrepreneurs set up a firm and produce a good that enters symmetrically in the utility function of consumers, generating the demand presented above. We assume that entrepreneurs need to pay in every period a fixed cost in order to run their business. This fixed cost takes the form of hiring a number of workers κ , on top of the productive labor demand in equation (5). The cost κ is the source of heterogeneity and determines the career choice.

The idea is that in the day-by-day running of the firm, entrepreneurs face options and have to take decisions. In order to take the right decision two types of abilities are required. The first is managerial talent: good entrepreneurs are better able to solve problems and therefore make larger profits. The second type of ability is related to the entrepreneurial environment. Given a certain level of managerial talent, entrepreneurs with a deeper knowledge of the entrepreneurial environment are able to take better decisions. We think of the entrepreneurial environment as the set of factors that shape the economy primarily (though not uniquely) through its *demand* side. Examples are cultural factors, language, tastes, but also regulations (e.g. different procedures to start up businesses) or any other factor that does not directly affect technology. Therefore, it is quite natural to think that entrepreneurial environments differ across countries and that entrepreneurs that have "local" knowledge have an advantage vis-à-vis entrepreneurs who do not have this knowledge.

We model these two types of entrepreneurial abilities (talent and local knowledge) by assuming that, in each period t, agents need to take two actions. In each case the "right action" is a number in the real line:

$$r_t \in \mathbb{R}$$
$$\mu_t \in \mathbb{R}$$

Both r_t and μ_t are random variables. In order to guess r_t , managerial talent is required; in order

to guess μ_t , it is useful to have local knowledge.²²

Entrepreneurs do not know the precise value of r_t and μ_t and take decisions based on their available information. The further away their action from the "right action", the larger the number of "unproductive" workers that are hired. That is, the expected number of "unproductive" workers κ is:

$$E(\kappa) = E (r_t - a)^2 + E (\mu_t - b)^2$$

We now specify the information set available to producers.

Information on r_t . We assume that all producers know that r_t is a normally distributed random variable, with independent draws over time, that has a certain known mean (whose value is irrelevant) and a variance V_r . In each period, before taking decisions, each entrepreneur receives an unbiased signal on r_t . The precision of the signal determines the ability of the entrepreneur. High-ability entrepreneurs receive more precise signals and make on average fewer mistakes.

If the precision of the signal received by the entrepreneur is τ , the variance of the posterior is $\frac{1}{V_r} + \tau$. Given that the optimal action is to choose *a* equal to the expectation of r_t , it is clear that:

$$E(\kappa) = Var(r_t) + Var(\mu_t) = \frac{1}{\frac{1}{V_r} + \tau} + Var(\mu_t)$$

Agents differ because they receive signals on r_t with different levels of precision. More talented entrepreneurs have larger precision (lower variance) and they expect to take, on average, more correct decisions. Consequently, they expect to hire fewer "unproductive" workers: the number of "unproductive" workers is decreasing in the talent of the individual.

Information on μ_t . Agents do not receive signals on the value of μ_t . Rather, they know that it evolves according to the following process:

$$\mu_t = \mu + u_t$$

where μ is a country-specific constant and u_t is an individual-specific white noise disturbance with zero mean and variance equal to σ_u . Agents take decisions before the realization of the shock u_t . We assume that domestic producers know μ and therefore they are only left with the residual uncertainty implied by the presence of the shock u_t . Therefore, due to the uncertainty on μ_t they hire, on average, only σ_u "unproductive" worker as, from their perspective, $Var(\mu_t) = Var(u_t) = \sigma_u$.

Foreigners, instead, do not know the exact value of μ and have to learn it by observing its realizations over time. They have a prior on it, with a certain precision P_t .²³ The variance of

 $^{^{22}}$ The assumption that the two decisions are separated and that each requires a different type of ability is a useful analytical simplification. It is possible to specify a model where a single decision is taken at the cost of getting a much more involved learning process without gaining further insights.

²³Notice that the prior is, in our model, a measure of the distance between entrepreneurial environments, as it tells how much foreign agents know about the domestic environment, *before starting the learning process*.

the beliefs of foreign producers on μ_t is $\frac{1}{P_t} + \sigma_u$. Consequently,

$$E(\kappa) = Var(r_t) + Var(\mu_t) = \frac{1}{\frac{1}{V_r} + \tau} + \sigma_u + \frac{1}{P_t}$$

At the end of each period agents learn the realization of μ_t and update their beliefs. As u_t has variance equal to σ_u , after having observed the realization of μ_t for s periods the precision is:

$$P_{t+s} = P_t + s \times \frac{1}{\sigma_u}$$

Over time foreign producers learn the exact value of μ , as their precision becomes infinite, and the expected number of "unproductive" workers to be hired because of the imperfect knowledge of the entrepreneurial environment converges to σ_u , which is the expected amount hired also by domestic entrepreneurs.

We index talent by ϕ and label $Var(r_t) + Var(u_t) = \frac{1}{\frac{1}{V_r} + \tau} + \sigma_u \equiv \phi$, with ϕ being distributed according to some CDF $F(\phi)$. This implies that:

$$E(\kappa) = Var(r_t) + Var(\mu_t) = Var(r_t) + Var(u_t) + \frac{1}{P_t} = \phi + \frac{1}{P_t}$$

In order to have a suitable benchmark when we allow for cross-border activity, the next section solves for the closed economy equilibrium.²⁴ In this case, as domestic producers know how to deal with the entrepreneurial environment (i.e. they have infinite precision on μ), we have that $E(\kappa) = Var(r_t) + Var(u_t) = \phi$. Therefore the expected net benefits and total labor demand of a firm run by an entrepreneur with talent ϕ are given respectively by:

$$E[\pi(\phi)] = \frac{Y}{w} - \phi w$$

$$L(\phi) = \frac{Y}{w^2} + \phi = z + \phi$$

It is useful to notice that:

$$E[\pi(\phi)] = (z - \phi) w$$

As standard, we assume that no individual can be entrepreneur and worker at the same time.

5 Closed Economy Equilibrium

At the aggregate level the only relevant price is the wage rate. Given a certain wage, agents are going to choose to become entrepreneurs if and only if:²⁵

$$w \le E[\pi(\phi)] \quad \Leftrightarrow \quad \phi \le z - 1$$

 $^{^{24}}$ Section 6 solves for the open economy.

 $^{^{25}}$ We assume the existence of perfect capital markets, so that only the expected profits (and not the actual profits) are relevant.

Assuming a continuum of agents of mass one, the mass of agents that will choose to be entrepreneurs is F(z-1). Hence, labor supply and demand are:

$$L_{S}(z) = 1 - F(z - 1)$$
(6)

$$L_D(z) = F(z-1) \times z + \int_0^{z-1} \phi dF(\phi)$$
(7)

Notice that en lieu of expressing supply and demand as a function of prices, we express them as a function of firms' productive labor demand (z). We do so for simplicity; z is a monotonously decreasing function of the wage rate, and expressing everything in terms of z simplify matters. An increase in z means that labor has become relatively cheaper relative to Y (thus labor demand goes up and supply down), and it is therefore convenient to think of z as of how cheap labor is.

Equilibrium is attained when (1) career choices (being an entrepreneur or not) are optimally taken; (2) the labor market clears (labor demand equals the mass of workers); (3) aggregate demand equals the total income generated in the economy. We will refer to the last two conditions as to the labor market equilibrium and the goods market equilibrium.

5.1 Labor Market

In what follows we are going to maintain the hypothesis that ϕ (the inverse of talent) is uniformly distributed in [0, 1].²⁶ Hence, from (6), labor supply reads as follows:

$$L_S(z) = \min\{\max[2-z,0],1\}$$

and from (7) labor demand is given by:

$$L_D(z) = \begin{cases} 0 & \text{If } z < 1\\ \frac{1}{2}(z-1)(3z-1) & \text{If } 1 \le z \le 2\\ z + \frac{1}{2} & \text{If } 2 < z \end{cases}$$

In equilibrium it must be that $1 \le z \le 2$, because if 2 < z labor supply is zero while if z < 1 labor demand is zero. Therefore, the condition $L_D = L_S$ boils down to:

$$\frac{3}{2}z^2 - z - \frac{3}{2} = 0$$

There is a unique positive solution to this equation, whose value is denoted by Z_A (specifically $Z_A = \frac{1}{3} + \frac{1}{3}\sqrt{10}$, which does not mean anything by itself). What is interesting is that z (how cheap is labor relative to the output of the economy) is the only variable that affects the labor market. This property will turn out to be very useful later on.

 $^{^{26}\}mathrm{Appendix}$ C solves the model using a general CDF.

5.2 Goods market

In equilibrium, aggregate demand Y must be equal to the total output of the economy. Total output per firm equals the sum of gross profits $(\pi = \frac{Y}{w} = z \times w)$ and the wage bill $(z \times w)$,²⁷ hence it equals 2zw. There are F(z-1) firms, thus the goods market equilibrium condition reads as follows:

$$Y = 2F\left(z-1\right)zw$$

and therefore, any equilibrium price needs to be such that w = 2F(z-1). Given the assumption that the distribution of talent is uniform: $\phi \sim U(0, 1)$:

$$w = 2\left(z - 1\right)$$

Thus, z (how cheap it is labor) completely characterizes the equilibrium and it is pinned down solely by the labor market clearing condition. For later reference it is useful to observe that, given our assumptions, the equilibrium values of wages and income are respectively $w_A = 2(Z_A - 1) = \frac{2}{3}\sqrt{10}$ and $Y_A = 4Z_A(Z_A - 1)^2 = \frac{40}{27}(1 + \sqrt{10})$.

6 The open economy

We now turn to a world where entrepreneurs are allowed to set up firms abroad,²⁸ i.e. engage in FDI.²⁹ In this setting, entrepreneurial ability is country-specific. Each country is assumed to be characterized by a different entrepreneurial environment and talent is defined as the ability to deal with the *domestic* environment. Thus, in order to successfully set up a firm abroad, entrepreneurs need to *learn* how the foreign entrepreneurial environment works. As explained in section 4.2, we model the learning cost as an extra amount of "unproductive" workers to be hired in order to run the firm properly. The expected number of "unproductive" workers is, in each period, equal to $\sigma_u + \frac{1}{P_0 + s \times \frac{1}{\sigma_u}}$ where s is the time elapsed since the foreign entrepreneur started producing abroad and $P_s = P_0 + s \times \frac{1}{\sigma_u}$ is the precision after observing the realization of μ_t for s periods. We denote the expected present value of the learning cost of a firm that starts producing abroad (and never leaves the foreign market) as $\overline{C}(\beta, P_0, \sigma_u) \equiv \sum_{s=0}^{\infty} \beta^s \frac{1}{P_0 + s \times \frac{1}{\sigma_u}}$.³⁰

Finally, we assume that entrepreneurs who set up a plant abroad incur in an additional cost that requires hiring an extra amount γ of workers. This assumption captures the idea that

²⁷Gross profits are then split between entrepreneurs' income and the fixed costs.

 $^{^{28}}$ We rule out trade. For a model of FDI and trade see Helpman, Melitz and Yeaple (2003).

²⁹We focus on "horizontal" FDI, i.e. investments aiming at establishing production facilities in a foreign country in order to serve the local market by making use of the local workforce. In other words, we restrict to goods that need to be produced in the same geographic location where they are consumed. One can think either of firms providing services, or firms facing substantial trade costs. As discussed in section 2.1, there is a general consensus that the overwhelming proportion of FDI is horizontal rather than vertical. We rule out both the licensing alternative (on this see Ethier (1986), Horstmann and Markusen (1987), and Ethier and Markusen (1996)) and "vertical" FDI, in which the production process is fragmented across countries (on this see Helpman (1984), Helpman (1985), Markusen (2002, Ch. 9)).

³⁰Notice that, provided that $\beta < 1$, $P_0 > 0$ and σ_u is finite, the expected present value of the learning cost is a finite number. To see why, note that $\sum_{s=0}^{\infty} \beta^s \frac{1}{P_0 + s \times \frac{1}{\sigma_u}} < \sum_{s=0}^{\infty} \beta^s < \infty$ because starting from some finite s each element of the first series is smaller than the corresponding term of the second (converging) series.

foreign entrepreneurs are less efficient than domestic ones (with the same level of talent) due, for example, to the difficulty of monitoring workers abroad. Thus, expected per period profits are $E[\pi_t(\phi)] = \left(z - \phi - \gamma - \frac{1}{P_0 + s \times \frac{1}{\sigma_u}}\right) w.$

6.1 Individual decisions

We focus on steady state analysis and characterize the values of the three possible career choices that each individual faces:

- 1. Be a worker.
- 2. Be a domestic entrepreneur.
- 3. Be a multinational entrepreneur.

In steady state, the value of being respectively a worker and a domestic entrepreneur, denoted by V_W and V_N , is given by:

$$V_W = w \frac{1}{1-\beta} \tag{8}$$

$$V_N = (z - \phi) w \frac{1}{1 - \beta}$$

$$\tag{9}$$

where β is the subjective discount factor. The above asset equations simply state that the values are equal to the present discounted values of the future streams of wages and net profits. The value of being a multinational entrepreneur and having already learned how the foreign country works, is:

$$V_X = V_N + \frac{1}{1 - \beta} \left(\widetilde{z} - \phi - \widetilde{\gamma} \right) \widetilde{w}$$
(10)

where the variables with a tilde refer to the foreign country. In equation (10) the first term of the right hand side is simply the value of being a domestic entrepreneur while the second term is the value of being a multinational "educated" entrepreneur, thus it accounts for the extra fixed cost $\tilde{\gamma}$ associated to FDI and not for the learning cost.

Finally, the value of becoming a multinational entrepreneur that still has to bear the learning cost in order to set up a firm abroad is:

$$V_{BX} = V_X - \overline{C}(\beta, P_0, \sigma_u)\widetilde{w}$$

= $V_N + \frac{1}{1 - \beta} \left(\widetilde{z} - \widetilde{\phi} - \gamma\right)\widetilde{w} - \overline{C}(\beta, P_0, \sigma_u)\widetilde{w}$

Where $\overline{C}(\beta, P_0, \sigma_u)\widetilde{w}$ is the present discounted value of the learning cost. Given that the cost decreases over time, no agent will produce abroad for some time only: if an agent sets up a firm abroad, it is forever.

6.1.1 Steady State Decisions

Using the above asset equations, we first look at the decision of individuals who have *not* been FDI entrepreneurs in the past and have therefore not learned yet how the entrepreneurial environment works in the foreign country. Then, we will analyze the decisions of the agents who already know how to deal with the foreign country environment.

- Individuals who have *not* engaged in FDI,
 - will choose to be workers if and only if³¹

$$z-1 \leq \phi$$

- will choose to be domestic entrepreneurs if and only if:

$$\widetilde{z} - \widetilde{\gamma} - (1 - \beta)\overline{C}(\beta, P_0, \sigma_u) \le \phi \le z - 1$$

- and will change their mind and set up a firm abroad if and only if

$$\phi \le (\widetilde{z} - \widetilde{\gamma}) - (1 - \beta)\overline{C}(\beta, P_0, \sigma_u)$$

- Individuals who have been running a firm abroad in the past (or that for any circumstances do not need to pay the learning cost) have a different decision scheme.
 - They become workers if and only if:

$$z - 1 \le \phi$$

- They become domestic entrepreneurs if and only if

$$\widetilde{z} - \widetilde{\gamma} \le \phi \le z - 1$$

- And continue being multinational entrepreneurs if and only if

$$\phi \le \widetilde{z} - \widetilde{\gamma}$$

Notice that agents with $\phi \in [(\tilde{z} - \tilde{\gamma}) - (1 - \beta)\overline{C}(\beta, P_0, \sigma_u), \tilde{z} - \tilde{\gamma}]$ will be engaging in FDI only if they have been doing so in the past, but will be domestic entrepreneurs otherwise.

If in steady state any agent who *could* be a multinational entrepreneur *is* actually one, then individuals split up among the three categories according to the following rules:

³¹The actual restriction is $\max\left\{z-1, (z-1)\frac{w}{w+\tilde{w}} + (\tilde{z}-\tilde{\gamma}-(1-\beta)\overline{C}(\beta,P_0,\sigma_u))\frac{\tilde{w}}{w+\tilde{w}}\right\} \leq \phi$. However, in what follows we will focus on the case where the set of domestic entrepreneurs is non-empty for any level of the learning cost. This happens whenever $\tilde{z}-z \leq \tilde{\gamma}-1$ (which simplifies to $1 \leq \tilde{\gamma}$ under symmetry). In this case, it is easy to see that the above condition boils down to the one presented in the text.

$$\begin{aligned} z-1 \leq \phi & (\text{Workers}) \\ -(1-\beta)\overline{C}(\beta,P_0,\sigma_u) \leq \phi \leq z-1 & (\text{Domestic entrepreneurs}) \\ \phi \leq \widetilde{z} - \widetilde{\gamma} - (1-\beta)\overline{C}(\beta,P_0,\sigma_u) & (\text{Exporting entrepreneurs}) \end{aligned}$$

In order to simplify notation we define a variable C capturing the discounted value of the future stream of learning costs: $C = (1 - \beta)\overline{C}(\beta, P_0, \sigma_u)$.

Notice that in this model individuals live forever. For this reason, in steady state there are no flows in and out of the different states. However, all our results carry over if we extend the model and assume an overlapping generations structure where individuals die with at a certain rate and newborns enter (at the same rate) with a random level of talent.

We now turn to the analysis of the labor market equilibrium and of the goods market equilibrium.

6.2 Labor market

 $\widetilde{z} - \widetilde{\gamma}$

In a symmetric steady state countries have the same aggregate income Y and the same wage rate w. The labor supply, in each country, is given by the total number of individuals who choose not to be entrepreneurs:

$$L_S(z) = 1 - F(z - 1)$$

The labor demand is given, in each country, by the sum of (i) the demand for labor of national entrepreneurs (ii) and the demand for labor of foreign entrepreneurs operating in the country. As in steady state the learning cost has been already paid, the labor demand reads as follows:

$$L_D(z) = \int_0^{(z-\gamma-C)} (z+\gamma+\phi) \, dF(\phi) + \int_0^{z-1} (z+\phi) \, dF(\phi)$$

= $[z+\gamma+E(\phi|\phi \le (z-(\gamma+C)))] F(z-(\gamma+C)) + [z+E(\phi|\phi \le (z-1))] F(z-1)$

where $\int_0^{(z-\gamma-C)} (z+\phi+\gamma)$ is the labor demand of foreign producers that we denote by L_D^f , and the second term $\int_0^{z-1} (z+\phi) dF(\phi)$ is the labor demand of domestic producers, denoted by L_D^d .

Notice that a drop in C (lower learning costs) affects the demand for labor via two margins. On the one side, a lower C induces a larger amount of foreigners to engage in FDI, thus increasing the domestic labor demand (*extensive* margin). On the other side, the *marginal* foreign entrepreneurs (and *only* the marginals) need to hire less workers (*intensive* margin) because the fixed cost C is lower.³² However, the net effect on labor demand of a decrease in C is clearly positive.³³ Recalling that talent is uniformly distributed ($\phi \sim U(0, 1)$) and realizing that the

 $^{^{32}}$ Notice that a *larger* C would affect labor demand only through the extensive margin because in that case a lower amount of foreigners would engage in FDI, leaving unaltered the labor demand of those who keep being multinationals.

³³The fixed cost γ affects the demand for labor as well, not only by changing the proportion of foreign agents who decide to produce in the home country (very much as the learning cost C), but also by affecting the labor demand of *all* foreign producers (and not only the marginals).

number of entrepreneurs that demand labor is bounded between zero and one:

$$L_{D} = \left[\max\left\{\min\left\{\left(z - (\gamma + C)\right), 1\right\}, 0\right\}\right](z + \gamma) + \frac{1}{2}\left[\max\left\{\min\left\{\left(z - (\gamma + C)\right), 1\right\}, 0\right\}\right]^{2} + \left[\max\left\{\min\left\{\left(z - 1\right), 1\right\}, 0\right\}\right]z + \frac{1}{2}\left[\max\left\{\min\left(z - 1\right), 1\right\}, 0\right\}\right]^{2}$$

With

$$\begin{split} L_D^f &= & \left[\max \left\{ \min \left\{ \left(z - \left(\gamma + C \right) \right), 1 \right\}, 0 \right\} \right] (z + \gamma) + \\ &+ \frac{1}{2} \left[\max \left\{ \min \left\{ \left(z - \left(\gamma + C \right) \right), 1 \right\}, 0 \right\} \right]^2 \right] \\ L_D^d &= & \left[\max \left\{ \min \left\{ \left(z - 1 \right), 1 \right\}, 0 \right\} \right] z + \frac{1}{2} \left[\max \left\{ \left\{ \min \left(z - 1 \right), 1 \right\}, 0 \right\} \right]^2 \right] \end{split}$$

Thus, domestic and foreign labor demands are given by:

$$L_D^d = \begin{cases} 0 & \text{If } z < 1\\ \frac{1}{2} (z-1) (3z-1) & \text{If } 1 \le z \le 2\\ z + \frac{1}{2} & \text{If } 2 < z \end{cases}$$
(11)

and

$$L_{D}^{f} = \begin{cases} 0 & \text{If } z < \gamma + C \\ \frac{1}{2} (z - \gamma - C) (3z + \gamma - C) & \text{If } \gamma + C \le z \le 1 + \gamma + C \\ z + \gamma + \frac{1}{2} & \text{If } 1 + \gamma + C < z \end{cases}$$
(12)

Total labor demand $(L_D = L_D^f + L_D^d)$ is increasing in z (how cheap labor is). The domestic labor demand is larger than the foreign if labor is relatively expensive (low z), while if wages are low (high z) the foreign labor demand is larger, because foreign firms need to have larger staffs to operate ($\gamma > 0$).

When $z \ge 2$ labor is so cheap that all domestic agents want to be entrepreneurs, at least locally. This, of course, cannot happen in equilibrium. When $z \ge 1 + \gamma + C$ labor is even cheaper (recall that $\gamma > 1$) and therefore everybody would like to engage in FDI, which again cannot happen in equilibrium.

Let us now turn to what *can* happen in equilibrium. Given our assumption that the cost of operating abroad is large enough ($\gamma > 1$), the shape of total labor demand depends on the size of $\gamma + C$.

We first analyze the case of a relatively high $(\gamma + C > Z_A)$. Notice that, in this case, whenever z is lower than $\gamma + C$ (labor is very expensive) the total labor demand equals the domestic labor demand, which is equal to the autarchic labor demand Z_A . Thus, not surprisingly, for relatively high fixed costs (and relatively high means precisely that $\gamma + C > Z_A$) the economy is *de facto* in autarchy. This can be seen in figure 2 (panel (a)). Foreign entrepreneurs demand labor in the home country only if it is very cheap, $z > \gamma + C$, so their presence has no effect in equilibrium.

We now turn to the case where the costs of opening a plant abroad are not so high $(\gamma + C < Z_A)$.

The total labor demand is now, in the relevant range, the sum of both domestic and foreign demand, as in panel (b) of figure 2. In that case, in equilibrium, both domestic and foreign entrepreneurs hire labor in the home country.

Since in equilibrium it can not happen neither than $z > 1 + \gamma + C$ (because then everybody would want to be a multinational entrepreneur) nor that z > 2 (because then everybody would want to be a domestic entrepreneur), the equilibrium condition reads as follows:

$$L_{S} = L_{D}$$

(2-z) = $\frac{1}{2}(z-1)(3z-1) + \frac{1}{2}(z-\gamma-C)(3z+\gamma-C)$

The above equation is satisfied for a unique value of z that we will denote $Z_T(\gamma, C)$:

$$Z_T(\gamma, C) = \frac{1}{6} \left((1 + \gamma + 2C) + \sqrt{(1 + \gamma + 2c)^2 + 6(\gamma - C)(\gamma + C) + 18} \right)$$

The value of z in equilibrium is then:

$$z^* = \begin{cases} Z_T(\gamma, C) & \text{If } \gamma + C \le Z_A \\ Z_A & \text{Otherwise} \end{cases}$$

Notice that when foreign producers actually hire in the home market, the labor demand cannot be smaller than in autarchy (see again figure 2), while labor supply is not affected by the possibility of cross-border investments. Thus it is clear that, in any equilibrium with multinational entrepreneurs (i.e. whenever $\gamma + C < Z_A$), (1) labor is relatively more expensive $(Z_A \ge Z_T(\gamma, C))$ and consequently (2) the number of workers is larger (the number of entrepreneurs smaller) than in autarchy.

The mass of agents that become entrepreneurs is smaller in each country if, in equilibrium, some (high-ability) agents invest across borders. This, anyway, does not mean that the number of firms that sells to consumers is going to be smaller, because entrepreneurs from both countries serve them.

6.3 Goods Market

In equilibrium, in each country total production must be equal to the income of its inhabitants (earned either at home or abroad), or equivalently, the income generated in each country (independently of the country of the earner) has to be equal to the total production, i.e.:

$$Y = F(z-1) 2zw + F(z-\gamma - C) 2zw$$

which implies that:

$$w = 2 \times [F(z-1) + F(z-\gamma - C)]$$

Thus, the wage is a linear function of the number of entrepreneurs that operate in the country.

6.4 The Distributional effects of Globalization

In this section we analyze the implications of the model concerning the distributional effects of globalization. We compare the steady states of two worlds with different values of C (in the introduction we deemed them as the *Globalized Universe* (low C) and the *National Universe* (high C)) and ask in which world an individual with a certain ϕ would choose to be sent.

In the appendix we prove that, in equilibrium, the following statements are true:

$$\frac{\partial z\left(C\right)}{\partial C} > 0 \tag{13}$$

$$\frac{\partial w\left(C\right)}{\partial C} < 0 \tag{14}$$

$$\frac{\partial Y(C)}{\partial C} < 0 \tag{15}$$

$$\frac{\partial \pi}{\partial C} = \frac{\partial \left[Y\left(C\right)/w\left(C\right)\right]}{\partial C} > 0 \tag{16}$$

It is not surprising that a larger learning cost determines a smaller equilibrium number of foreign entrepreneurs (equation (13)). Less obvious is that this will reduce wages (equation (14)) and increase the gross profits of firms (equation (16)). We plot the above functions for a particular value of γ ($\gamma = 1.1$) in figure 3.

A larger C reduces the wage rate because it induces some foreign producers not to engage in FDI. This reduces the domestic labor demand and the wage rate. In turn, this induces a positive proportion of (high-ability) domestic workers to become (local) entrepreneurs, thus partially offsetting the drop in labor demand. Despite this, the net effect on the wage rate is negative. Thus, workers prefer to live in a more open universe, i.e. one where entrepreneurial environments are more similar.

A larger C also implies that, in equilibrium, both the average and the marginal firms are less efficient. First, some very efficient foreign producers are expelled from the local market. Second, a positive mass of domestic workers with low entrepreneurial talent turn to the entrepreneurial activity, further lowering aggregate productivity.

Moreover, in case of a large differences in the entrepreneurial environment (high C), aggregate demand (Y) is lower, so that gross profits $\left(\frac{Y}{w}\right)$ could in principle be either larger or smaller, as both revenues and costs are lower. Nevertheless, the effect through wages is larger, meaning that a larger C induces larger gross profits. The effect on net profits varies across the different types of entrepreneurs. The net profits of *local* producers (i.e. of agents that keep being local and do not *start* engaging in FDI) are larger the larger C, because the fixed cost ϕ does not depend on the learning cost, and, being wages lower, the total fixed cost ϕw goes down. Thus, for local entrepreneurs the increase in net profits is larger than the increase in gross profits.

As to foreign producers, the effect of this would-be exercise of considering a larger C is clear. A very talented individual facing the choice of which world to choose, would rather live in the world with the lowest level of C. The reason is that, once landed in this world, this individual would like to engage in FDI and would therefore suffer from a large C, because the negative direct effect of C on net profits more than offsets the increase in gross profits. Summing up, workers and talented entrepreneurs that engage in FDI prefer to live in a world with the lowest possible level of C. Differently, local producers prefer to live in a world with a high value of C.³⁴

Finally, note that with lower cross-the-border investments, even if the mass of agents that become entrepreneurs in each country increases, the mass of varieties available goes down. Dixit-Stiglitz love for variety implies then that lower "openness" lowers efficiency. Notice that this decrease in variety is a direct consequence of the entrepreneurs being less productive (and thus requiring more workers) in an economy with high learning costs. Openness (low learning costs) is (behind the veil of ignorance) welfare improving because it improves the allocation of talent.

7 Concluding remarks

This paper first presents empirical evidence on the positive effect of cross-country proximity in "entrepreneurial environments" on bilateral FDI flows. By exploiting the *OECD International Direct Investment Statistics* and data on nationwide regulation levels from the OECD and the World Bank, we find evidence that smaller differences in regulations across countries tend to be associated with larger bilateral flows of FDI, even controlling for the *level* of regulation, for countries fixed effects and for time effects, in the context of a standard gravity model.

Motivated by this evidence, we build a general equilibrium model that – while consistent with the main stylized facts about FDI – allows to study the distributional effects of globalization. In the model agents are heterogeneous and differ both in their ability to be entrepreneurs or workers and their nationality. Entrepreneurs *may* set up a firm abroad, i.e. engage in FDI. If they do so they incur in two additional costs, one of which is the cost of *learning* how the foreign environment works. In this framework, globalization fosters FDI and improves the allocation of talents in the economy boosting wages, output, and productivity.

The mechanism is as follows. Only the more able entrepreneurs engage in FDI, and their fraction grows larger the "easier" it is to set up a firm abroad, i.e. the more similar the domestic and foreign entrepreneurial environments. This, in turn, increases the demand for domestic labor, output and wages. As a consequence, the minimum ability level needed to become an entrepreneur goes up. This implies that the size of the pool of entrepreneurs goes down while its composition changes: a greater proportion engages in FDI. Hence, globalization improves the allocation of talent of the economy because the increase in the wage rate dissuades low-ability people from becoming entrepreneurs. At the same time, even if less people opt for an entrepreneurial career, a larger fraction serve clients abroad, implying that the variety of products that costumers may acquire increases.

The model implies that high- and low-ability agents are better off in a globalized universe, i.e. in a world where learning costs are lower. The reason is that low-ability agents, who decide to become workers, earn higher wages in a globalized environment, whereas high-ability agents can exploit larger investment opportunities. Medium-ability agents, differently, prefer to live in

³⁴The four functions object of this discussion (wage rate, local entrepreneurs profits, multinational entrepreneurs profits, and profits of becoming a multinational entrepreneur) are plotted in figure 4 for the agent with $\phi = \frac{1}{2}$ (with $\gamma = 1.1$).

a national universe where they are sheltered from foreign competition.

References

- Baldwin, Richard E., Henrik Braconier, and Rikard Forslid (2005), "Multinationals, Endogenous Growth and Technological Spillovers: Theory and Evidence", *Review of International Economics*, Volume 13 Page 945 - November 2005
- [2] Barba-Navaretti, G. and Anthony Venables, (2004), "Multinational firms in the world economy", with and F. Barry, K. Ekholm, A. Falzoni, J. Haaland, K-H. Midelfart and A. Turrini, Princeton University Press.
- [3] Brainard, S. Lael. (1993), "An empirical assessment of the factor proportions explanation of multinational sales", NBER Working Paper No. 4583.
- [4] Brainard, S. Lael. (1997), "An empirical assessment of the proximity-concentration trade-off between multinational sales and trade", American Economic Review, 87, 520-544.
- [5] Deardorff, A. V. (1998), "Determinants of bilateral trade: Does gravity work in a neoclassical world?", in J. A. Frankel (ed.), *The Regionalization of the World Economy*, Chicago: Chicago University Press.
- [6] Ekholm, K., (1998), "Proximity advantages, scale economies, and the location of production", in Braunerhjelm, P. and K. Ekholm (eds.) The Geography of Multinational Firms, Dordrecht: Kluwer Academic Publishers.
- [7] Ethier, Wilfred J., (1986), "The Multinational Firm." The Quarterly Journal of Economics, 101:805—33.
- [8] Ethier, Wilfred J. and James R. Markusen, (1996), "Multinational Firms, Technology Diffusion and Trade." Journal of International Economics 41:1—28.
- [9] Frankel, Jeffrey, Ernesto Stein and Shang-Jin Wei, (1995), "Trade Blocs and the Americas: The Natural, The Unnatural, and the Super-Natural", *Journal of Development Economics*, Vol. 47, Issue 1, June 1995, Pages 61—95.
- [10] Frankel, Jeffrey, and Shang-Jin Wei, (1998), "Open Regionalism in a World of Continental Blocs", NBER Working Paper No 5272.
- [11] Hassler, John and J.V. Rodríguez Mora, (2000), "Intelligence, Social Mobility, and Growth", *The American Economic Review*, Vol. 90(4), September, pp. 888-908.
- [12] Helpman, Elhanan, (1984), "Simple theory of international trade with multinational corporations", Journal of Political Economy, 92, 451-471.
- [13] Helpman, Elhanan, (1985), "Multinational Corporations and Trade Structure." *The Review* of *Economic Studies* 52:443—57.

- [14] Helpman, Elhanan, M.J. Melitz and S.R. Yeaple, (2004), "Export vs. FDI with heterogeneous firms", American Economic Review, Vol. 94(1), March, pp. 300-316.
- [15] Horst, T., (1972), "Firm and Industry Determinants of the Decision to Invest Abroad: An Empirical Study", *Review of Economics and Statistics* 54, 258-266.
- [16] Horstmann, Ignatius and James R. Markusen, (1987), "Licensing versus Direct Investment: A Model of Internalization by the Multinational Enterprise." *Canadian Journal of Economics* 20:464-81.
- [17] Horstmann, Ignatius and James R. Markusen, (1992), "Endogenous market structure in international trade (Natura Facit saltum)", Journal of International Economics, 32, 109-129.
- [18] Javorcik, Beata Smarzynska, (1992), "Does Foreign Direct Investment Increase the Productivity of Domestic Firms? In Search of Spillovers Through Backward Linkages", American Economic Review, Volume 94, Number 3, 1 June 2004, pp. 605-627(23).
- [19] Keller, Wolfgang and Stephen R. Yeaple, (2003), "Multinational Enterprises, International Trade and Productivity Growth: Firm Level Evidence From The United States", NBER Working Paper 9504.
- [20] Lipsey, R. E., (2001), "Foreign Direct Investment and the Operations of Multinational Firms: Concepts, History and Data", NBER Working Paper 8665.
- [21] Lipsey, R. E., (2002), "Home and Host Country Effects of FDI", NBER Working Paper 9293.
- [22] Markusen, James R., (2002), Multinational Firms and the Theory of International Trade, MIT Press.
- [23] Markusen, J. R. and K. Maskus, (2001), "Multinational firms: Reconciling theory and evidence", in M. Blomström and L. S. Goldberg (eds.), *Topics in International Economics.* A Festschrift in Honor of Robert E. Lipsey, Chicago: Chicago University Press.
- [24] Markusen, J. R. and K. Maskus, (2002), "Discriminating among alternative theories of the multinational enterprise", *Review of International Economics*, 10, pp.694-707.
- [25] Melitz, Marc J., (2003), The Impact of Trade on Intra-Industry Reallocations and Aggregate Industry Productivity, *Econometrica*, Vol. 71(6), November, pp. 1695-1725.
- [26] Nicoletti, G., Stefano Scarpetta and Olivier Boylaud, (2000), "Summary indicators of product market regulation with an extension to employment protection legislation" Economics Department Working Papers No. 226.
- [27] Razin, A., Yona Rubinstein and Efraim Sadka, (2004), "Which Countries Export FDI, and How Much?", CEPR Working Paper No 4204.
- [28] Shatz, H. J., (2003), "Gravity, education, and economic development in multinational affiliate location", Journal of International trade and economic development, 12.1, 117-150.

- [29] Wooldridge, Jeffrey M. 2003. Introductory Econometrics. 2 ed. South-Western
- [30] World Bank, *Doing Business 2004*, available at http://rru.worldbank.org/ doingbusiness.

A Proofs

We now prove that the following statements are true:

$$\frac{\partial z (C)}{\partial C} > 0$$

$$\frac{\partial w (C)}{\partial C} < 0$$

$$\frac{\partial Y (C)}{\partial C} < 0$$

$$\frac{\partial Y (C)}{\partial C} < 0$$

$$\frac{\partial [Y (C) / w (C)]}{\partial C} > 0$$

Proof. First we prove that $\frac{\partial z(C)}{\partial C} > 0$

Recall that:

$$z(C) = \frac{(1+\gamma) + 2C + \sqrt{((1+\gamma) + 2C)^2 - 12\left(\frac{1}{2}\left(C^2 - \gamma^2\right) - \frac{3}{2}\right)}}{6}$$

then,

$$\frac{\partial z(C)}{\partial C} = \frac{1}{3} \left(1 + \frac{(1+\gamma - C)}{\sqrt{(1+\gamma + 2C)^2 - 12\left(\frac{1}{2}\left(C^2 - \gamma^2\right) - \frac{3}{2}\right)}} \right) > 0$$

since |

$$\begin{array}{rcl} \gamma & > & 1 \\ \gamma + C & < & Z_A \end{array}$$

which implies that C < 1 and $\gamma + 1 > 2$, therefore $\gamma + 1 - C > 0$.

Proof. Next, we prove that $\frac{\partial w(C)}{\partial C} < 0$, where

$$w = 4z - 2\left(\gamma + C + 1\right)$$

hence,

$$\frac{\partial w(C)}{\partial C} = 4 \frac{\partial z(C)}{\partial C} - 2
= \frac{4}{3} \left(\frac{(1+\gamma-C)}{\sqrt{(1+\gamma+2C)^2 - 12\left(\frac{1}{2}\left(C^2 - \gamma^2\right) - \frac{3}{2}\right)}} - \frac{1}{2} \right)$$
(17)

Which is negative if

$$(1 + \gamma - C) < \frac{1}{2} \sqrt{(1 + \gamma + 2C)^2 - 12\left(\frac{1}{2}(C^2 - \gamma^2) - \frac{3}{2}\right)} (1 + \gamma)^2 < 2(\gamma^2 - C^2) + 6 + \frac{4}{3}(1 + \gamma)C$$
(18)

Which is true since $(1 + \gamma)^2$ must be smaller than $(1 + Z_A)^2 = 5.6997$ and the right hand side is larger than 6.

Proof. Next, we prove that $\frac{\partial Y(C)}{\partial C} = \frac{\partial z(C)w(C)^2}{\partial C} < 0.$

$$\frac{\partial Y(C)}{\partial C} = \frac{\partial z(C)}{\partial C} w^2 + 2zw \frac{\partial w(C)}{\partial C}$$
$$= w \left(4z \frac{(1+\gamma-C)}{\sqrt{(1+\gamma+2C)^2 - 12\left(\frac{1}{2}\left(C^2 - \gamma^2\right) - \frac{3}{2}\right)}} - \frac{\partial z(C)}{\partial C} 2\left(\gamma+C+1\right) \right)$$

recalling that

$$z = \frac{(1+\gamma) + 2C + \sqrt{((1+\gamma) + 2C)^2 - 12\left(\frac{1}{2}\left(C^2 - \gamma^2\right) - \frac{3}{2}\right)}}{6}$$

then,

$$\begin{aligned} \frac{\partial Y\left(C\right)}{\partial C} &= w^2 \frac{\left(1+\gamma\right)+2C+\sqrt{\left(\left(1+\gamma\right)+2C\right)^2-12\left(\frac{1}{2}\left(C^2-\gamma^2\right)-\frac{3}{2}\right)}}{3\sqrt{\left(1+\gamma+2C\right)^2-12\left(\frac{1}{2}\left(C^2-\gamma^2\right)-\frac{3}{2}\right)}} \left(1+\gamma-C\right) - w \frac{\partial z\left(C\right)}{\partial C}^2\left(\gamma+C+1\right) \\ &= w \frac{2\left(\left(1+\gamma\right)^2-C^2\right)+2C\left(1+\gamma-C\right)}{3\sqrt{\left(1+\gamma+2C\right)^2-12\left(\frac{1}{2}\left(C^2-\gamma^2\right)-\frac{3}{2}\right)}} + w \frac{2\left(1+\gamma-C\right)}{3} - w \frac{\partial z\left(C\right)}{\partial C}^2\left(\gamma+C+1\right) \end{aligned}$$

and using again

$$\frac{\partial z(C)}{\partial C} = \frac{1}{3} \left(1 + \frac{(1+\gamma - C)}{\sqrt{(1+\gamma + 2C)^2 - 12\left(\frac{1}{2}\left(C^2 - \gamma^2\right) - \frac{3}{2}\right)}} \right)$$

we get that:

$$\begin{aligned} \frac{\partial Y\left(C\right)}{\partial C} &= w \frac{2\left(\left(1+\gamma\right)^2 - C^2\right) + 2C\left(1+\gamma - C\right)}{3\sqrt{\left(1+\gamma+2C\right)^2 - 12\left(\frac{1}{2}\left(C^2 - \gamma^2\right) - \frac{3}{2}\right)}} + w \frac{2\left(1+\gamma - C\right)}{3} \\ &- \frac{w}{3}\left(1 + \frac{\left(1+\gamma - C\right)}{\sqrt{\left(1+\gamma+2C\right)^2 - 12\left(\frac{1}{2}\left(C^2 - \gamma^2\right) - \frac{3}{2}\right)}}\right) 2\left(\gamma + C + 1\right) \\ &= w \frac{2}{3}C\left(\frac{\left(1+\gamma - C\right)}{\sqrt{\left(1+\gamma+2C\right)^2 - 12\left(\frac{1}{2}\left(C^2 - \gamma^2\right) - \frac{3}{2}\right)}} - 2\right) < 0\end{aligned}$$

the above expression is negative since, as shown above:

$$\frac{(1+\gamma-C)}{\sqrt{(1+\gamma+2C)^2 - 12\left(\frac{1}{2}\left(C^2 - \gamma^2\right) - \frac{3}{2}\right)}} < \frac{1}{2}$$

-	
L	
_	

Proof. Finally, we prove that $\frac{\partial [Y(C)/w(C)]}{\partial C} = \frac{\partial (z(C)w(C))}{\partial C} > 0$

$$\begin{aligned} \frac{\partial \left(z\left(C \right) w\left(C \right) \right)}{\partial C} &= \frac{\partial z\left(C \right)}{\partial C} w + \frac{\partial w\left(C \right)}{\partial C} z \\ &= 2z \left(\frac{1}{3} + \frac{4}{3} \frac{\left(1 + \gamma - C \right)}{\sqrt{\left(1 + \gamma + 2C \right)^2 - 12\left(\frac{1}{2} \left(C^2 - \gamma^2 \right) - \frac{3}{2} \right)}} \right) - \\ &- \frac{2}{3} \left(1 + \frac{\left(1 + \gamma - C \right)}{\sqrt{\left(1 + \gamma + 2C \right)^2 - 12\left(\frac{1}{2} \left(C^2 - \gamma^2 \right) - \frac{3}{2} \right)}} \right) (\gamma + C + 1) \end{aligned}$$

where being:

$$z = \frac{(1+\gamma) + 2C + \sqrt{\left((1+\gamma) + 2C\right)^2 - 12\left(\frac{1}{2}\left(C^2 - \gamma^2\right) - \frac{3}{2}\right)}}{6}$$

we get that

$$\begin{aligned} \frac{\partial \left(z\left(C\right)w\left(C\right)\right)}{\partial C} &= 4\frac{\left(1+\gamma\right)+2C+\sqrt{\left(\left(1+\gamma\right)+2C\right)^{2}-12\left(\frac{1}{2}\left(C^{2}-\gamma^{2}\right)-\frac{3}{2}\right)}}{9\sqrt{\left(1+\gamma+2C\right)^{2}-12\left(\frac{1}{2}\left(C^{2}-\gamma^{2}\right)-\frac{3}{2}\right)}}\right)} \left(1+\gamma-C\right) \\ &+ \frac{2}{3}z - \frac{2}{3}\left(1+\frac{\left(1+\gamma-C\right)}{\sqrt{\left(1+\gamma+2C\right)^{2}-12\left(\frac{1}{2}\left(C^{2}-\gamma^{2}\right)-\frac{3}{2}\right)}}\right)}\left(\gamma+C+1\right) \end{aligned}$$

hence,

$$\begin{aligned} \frac{\partial \left(z\left(C\right)w\left(C\right)\right)}{\partial C} &= \frac{1}{9}3C + \frac{1}{9}\left((1+\gamma) - C\right) \times \\ \times \left[\frac{\sqrt{\left((1+\gamma) + 2C\right)^2 + 12\left(\frac{1}{2}\left(\gamma^2 - C^2\right) + \frac{3}{2}\right)}}{(1+\gamma) - C} - \left(1 + \frac{2\left(1+\gamma - C\right)}{\sqrt{\left(1+\gamma + 2C\right)^2 - 12\left(\frac{1}{2}\left(C^2 - \gamma^2\right) - \frac{3}{2}\right)}}\right)\right] \end{aligned}$$

the above is negative since the expression in square brackets

$$\frac{\sqrt{\left((1+\gamma)+2C\right)^2+12\left(\frac{1}{2}\left(\gamma^2-C^2\right)+\frac{3}{2}\right)}}{(1+\gamma)-C} - \left(1+\frac{2\left(1+\gamma-C\right)}{\sqrt{\left(1+\gamma+2C\right)^2-12\left(\frac{1}{2}\left(C^2-\gamma^2\right)-\frac{3}{2}\right)}}\right) > 0$$

may be rewritten as follows

and it is clearly positive if
$$l = \frac{2(1+\gamma-C)}{\sqrt{(1+\gamma+2C)^2-12(\frac{1}{2}(C^2-\gamma^2)-\frac{3}{2})}} < \frac{1}{2}$$
, which holds true by (18).

B Figures and Tables

Australia	Netherlands
Belgium-Luxembourg	New Zealand
Canada	Norway
Denmark	Poland
Finland	Portugal
France	Spain
Germany	Sweden
Greece	Switzerland
Ireland	Turkey
Italy	United Kingdom
Japan	United States

Table 1: Countries

Table 2: Correlations between the OECD Regulation Subindexes

	\mathbf{PMR}	State	B. Ent.	В. Т&I	Ec. reg.	Ad. reg.	Inw.	Outw.
PMR	1							
State Control	0.86	1						
Bar. Entrep.	0.54	0.42	1					
Bar. T&I	0.76	0.45	0.03	1				
Econ. reg.	0.84	0.98	0.47	0.4	1			
Admin. reg.	0.49	0.36	0.97	0.01	0.38	1		
Inward reg.	0.85	0.89	0.79	0.37	0.90	0.73	1	
Outward reg.	0.82	0.52	0.1	0.99	0.47	0.08	0.40	1

Table 3: Variables of the World Bank dataset Doing Business 2004

Starting a Business	Number of procedures
	Average time spent during each procedure (in calendar days)
	Official cost of each procedure ($\%$ of income per capita)
	Paid-in minimum capital (as a percentage of income per capita)
Hiring and Firing	Difficulty of hiring index
0 0	Rigidity of hours index
	Difficulty of firing index
	Rigidity of employment (average of the three above)
	Firing costs (number of weeks)
Registering Property	Number of procedures
100glottor ting 1 roportig	Number of days
	Official cost (% of property value per capita)
Getting Credit	Cost to create and register collateral (% of income per capita) Index of legal rights of borrowers and lenders
	Index of credit information availability
	Coverage of public registry (borrowers per 1000 capita) Coverage of private registry (borrowers per 1000 capita)
Protecting Investors	Disclosure of ownership and financial information index
Enforcing Contracts	Number of procedures
	Number of days
	Official cost ($\%$ of the debt value)
Closing a Business	Number of years
	Official cost ($\%$ of estate)
	Recovery rate (cents on the dollar)

Dependent variable: FDI	Inflows as a s	hare of the ho	st country G	DP.				
Regulation Variables	Panel A:	Panel B:	Panel C:	Panel D:	Panel E:	Panel F:	Panel G:	Panel H:
Host Country EPL Level	000047	0000757	0000487	0000425	0000433	0000476	0000684	0000355
Source Country EPL Level	(.0001426 0001426 (000346)	(.0002966 0002966 (.0000439)	(.0003574 (.0000447)	(.00003306)	(00003336)	(.00003513 $(.0000447)$	(.0000437)	(.0000454)
Distance between regulations: Product Market	0002163							
Barriers to Trade and Investi	(curron) nent	0003356						
Barriers to Entrepreneurship		(anonnn.)	0003232					
State control			(eernnnn)	0001446				
Economic Regulation				(4000000.)	0002274			
Administrative Regulation					(sonnnn.)	0002177		
Overall outward-oriented reg	ulation					(seconon.)	0002832	
Overall inward-oriented regul	ation						(ROONNN)	000352
R^2 Observations	0.1505 3713	0.1535 3713	0.1538 3713	0.1511 3713	0.1521 3713	0.1526 3713	0.1519 3713	0.1543 0.1543 3713
Notes: The distance between re	gulations is m	easured as the	absolute value	e of the differe	nce between th	ie source and t	the host countr	y regulations.
The level of regulation is accou	nted for by a	time-varying n	ieasure of EPI	, (evaluated in	1990 and 199	8). All specific	cations include	the following
control variables: host country f	ixed-effects, h	ost and source	country GDP a	and populatior	ı, a time trend,	, distance betw	een main cities	l, latitude and
longitude of the source country;	common langu	lage dummy, E	U dummy, NAI	FTA dummy, c	ommon land be	orders dummy,	both in Asia F	acific dummy,
both in North America dummy.	Standard erre	ors in parenthe	sis.					

Table 4: **OECD Regulation variables.**

Dependent variable: FDI	Inflows as a sl	nare of the ho	ost country G	DP.					
Regulation Variables	Panel A:	Panel B:	Panel C:	Panel D:	Panel E:	Panel F:	Panel G:	Panel H:	Panel I:
Host Country EPL Level	0000615	0000574 (.0001559)	0000578 (.0001559)	0000545 (.000156)	000048 (.000156)	0000639 (.0001558)	0000482	0000491 (.0001559)	0000556 (.000156)
Source Country EPL Level	(.0000452)	(.0002807)	(.00002929)	0003136 (.0000438)	(.0000442)	(.00003014 $(.0000438)$	(.0003055)	(.000304)	(.0002903)
Distance between regulations: N. of procedures	: Starting a 0000328	Business							
Number of days		-4.82e-06							
Cost (% of income per capits	(1)	(00-200.7)	0000148						
Minimum capital (% of incor	ne per capita)		(0.406-00)	-1.32e-06 (1.23 $e-06$)					
Distance between regulations: Difficulty of hiring	: Hiring and	Firing			-3.95e-06 (1 85 <u>0.06</u>)				
Rigidity of hours						-8.67e-06			
Difficulty of firing							-6.45e-06		
Rigidity of employment								-6.61e-06 (2.53e-06)	
Firing costs (number of week	(S)								-3.34e-06 (1.69e-06)
R^2 Observations	$\begin{array}{c} 0.1504 \\ 3713 \end{array}$	$\begin{array}{c} 0.1508\\ 3713\end{array}$	$\begin{array}{c} 0.1512 \\ 3713 \end{array}$	0.1498 3713	0.1506 3713	0.1529 3713	$\begin{array}{c} 0.1512 \\ 3713 \end{array}$	0.1511 3713	$\underbrace{0.1504}_{3713}$
Notes: The distance betw	een regulations	is measured a	as the absolute	e value of the	difference betv	veen the source	e and the host	country regul	ations.
The level of regulation is	accounted for	by a time-vary	/ing measure o	of EPL (evalue	ated in 1990 a	nd 1998). All	specifications	include the fol	lowing
control variables: host cou	ıntry fixed-effec	ts, host and s	ource country	GDP and pop	ulation, a time	trend, distanc	e between mai	in cities, latitu	de and
longitude of the source cot both in North America du	intry; common immy. Standare	language dum d errors in par	my, EU dumm enthesis.	y, NAF'TA dui	nmy, common	land borders d	ummy, both in	. Asia Pacific d	ummy,

Table 5: World Bank Regulation variables: Starting a Business and Hiring and Firing

Table 6: V	Vorld Bank	Regulation	variables: <i>I</i>	Registering	Property a	nd Getting	Credit	
Dependent variable: FDI	Inflows as a	share of the h	ost country (GDP.				
Regulation Variables	Panel A:	Panel B:	Panel C:	Panel D:	Panel E:	Panel F:	Panel G:	Panel H:
Host Country EPL Level	0000548 (.000156)	0000553 (.000156)	0000512 (.0001561)	000053 (.000156)	0000569 (.0001559)	0000576 (.000156)	0000527 (.000156)	0000527 (.000156)
source Country EFL Level	000297	(.0000439)	0000448)	0003193 (.0000443)	(.0000438)	(.0000438)	0000454) (.0000454)	0003229 ($.0000438$)
Distance between regulations N. of procedures	:: Registerin 0000371 (.0000176)	ıg Property						
Number of days		-1.32e-06 (7.75e-07)						
Cost (% of property value p	er capita)		0000102 (.0000137)					
Distance between regulations Cost to create collateral ($\%$ c	:: <i>Getting C</i> of income per	J redit capita)		-2.75e-06 (6.20e-06)				
Legal rights index					0000607 (.0000231)			
Credit information index					~	0000603 (.0000446)		
Public registry coverage							-3.74e-07 (2.57 $e-07$)	
Private bureau coverage								-2.22e-07 (1.09e-07)
$\frac{R^2}{2}$.	0.1506	0.1502	0.1497	0.1496	0.1511	0.1500	0.1500	0.1505
Observations	3713	3713	3713	3713	3713	3713	3713	3713
Notes: The distance between The level of regulation is acco	regulations is 1 ounted for by a	measured as the time-varying m	absolute value reasure of EPL	of the difference (evaluated in]	te between the 1998).	source and the All specificati	host country re ons include the	gulations. following
control variables: host country	y fixed-effects, l	host and source	country GDP a	nd population,	a time trend, d	istance between	main cities, lat	itude and
longitude of the source country both in North America dumm	y; common lang w. Standard en	guage dummy, El rors in parenthes	U dummy, NAF sis.	'TA dummy, coi	mmon land borc	lers dummy, bo	th in Asia Pacifi	c dummy,

			f minon non				
Regulation Variables	Panel A:	Panel B:	Panel C:	Panel D:	Panel E:	Panel F:	Panel G:
Host Country EPL Level	0000454	0000515	0000497	0000517	0000518	0000514	0000511
Source Country EPL Level	(.0001301) - $.0003215$ (.0000439)	(6661000.) 0003281 (.000044)	(.00026710002671)	(00003189 - 0003189)	(.0002998 0002998 (.0000446)	(001100) 0003123 (.000044)	(00003178) 0003178 (.0000438)
Distance between regulations Disclosure index	: Protecting 0000673 (.0000473)	l Investors					
Distance between regulations N. of procedures	Enforcing	Contracts 0000232					
Number of days		(9.976-00)	-6.29e-07				
Cost ($\%$ of debt)			(1.44e-07)	.0000106 $(.0000106)$			
Distance between regulations. Number of years	: Closing a	Business			.0000918		
Cost ($\%$ of estate)					(6140000.)	-9.04e-06	
Recovery Rate (cents on the	dollar)					(00-960.6)	2.21e-06 (2.44e-06)
$\frac{R^2}{Observations}$	0.1500 3713	0.1508 3713	0.1539 3713	0.1498 3713	$\begin{array}{c} 0.1504 \\ 3713 \end{array}$	0.1497 3713	0.1497 3713
et a contract of the second se	ons is measured or by a time-va	as the absolut rying measure	ie value of the o of EPL (evalua	lifference betwe ted in 1990 and	en the source a 1 1998). All sp	nd the host cc ecifications inc	ountry regulatio

Table 8: "Beta" Co	pefficients.
--------------------	--------------

Variable	Obs	Mean	Std. Dev.	Beta Coeffs
Dependent variable				
FDI Inflows (share of host country GDP)	3713	0.00077	0.00232	
× · · · /				
Distance in regulation				
OECD variables				
Product Market Regulation	3713	0.62893	0.47054	-0.0439
State control	3713	1.09550	0.7763	-0.0484
Barriers to Entrepreneurship	3713	0.81336	0.56037	-0.0781
Barriers to Trade and Investment	3713	0.59838	0.61986	-0.0897
Economic Regulation	3713	0.89041	0.6468	-0.0634
Administrative Regulation	3713	1.0114	0.66697	-0.0626
Overall inward-oriented regulation	3713	0.85109	0.6123	-0.0747
Overall outward-oriented regulation	3713	0.55416	0.55855	-0.0847
Starting a business				
N. of procedures	3713	3.36386	2.5742	-0.0364
Number of days	3713	24.37	27.59	-0.0573
Cost (% of income per capita)	3713	8.533	7.8186	-0.0499
Minimum capital (% of income per capita)	3713	36.733	39.549	-0.0225
Hiring and firing				
Difficulty of hiring index	3713	29.380	22.694	-0.0386
Rigidity of hours index	3713	27.606	22.397	-0.0837
Difficulty of firing index	3713	19.504	15.955	-0.0444
Rigidity of employment (average of the three)	3713	24.197	16.947	-0.0483
Firing costs (number of weeks)	3713	29.046	28.587	-0.0412
Registering a property				
N. of procedures	3713	3.1023	2.4196	-0.0387
Number of days	3713	51.922	63.134	-0.0359
Official cost (% of property value per capita)	3713	3.8360	3.2295	-0.0142
Getting credit				
Cost of collateral (% of income per capita)	3713	7.0121	7.1774	-0.0085
Index of legal rights of borrowers and lenders	3713	2.6073	1.9637	-0.0514
Index of credit information availability	3713	1.1053	0.86101	-0.0224
Coverage of public registry	3713	132.3	204.81	-0.0330
Coverage of private registry	3713	460.01	349.15	-0.0334
Protecting investors				
Disclosure of ownership and financial info	3713	1.1664	0.9665	-0.0280
Enforcing contracts				
N. of procedures	3713	5.801	4.419	-0.0442
Number of days	3713	253.60	368.99	-0.1000
Official cost ($\%$ of debt value)	3713	5.1653	3.4790	0.0159
Closing a business				
Number of years	3713	1.0935	0.94187	0.0372
Official cost ($\%$ of estate)	3713	4.5354	4.303592	-0.0168
Recovery rate (cents on the dollar)	3713	21.225	16.0981	0.0153



Figure 2: Labor market equilibrium with large and small learning costs



Figure 3: Effects of openness (lower C) on gross profits, wages and output. ($\gamma = 1.1$).



Figure 4: Effects of openness (lower C) for the median agent ($\phi = \frac{1}{2}$) on wages (dotted line), net profits from domestic activity (diamond line), net profits from foreign activity (solid line), net profits from *starting* a foreign activity (thick solid line). ($\gamma = 1.1$).

C A two-sector model with tradable goods

The model described in the main text is highly stilyzed. Among the many simplifications, we have also assumed tradable goods away. This, however, implies that multinational entrepreneurs cannot consume the profits earned abroad. In this appendix we show that adding a perfectly competitive sector of tradable goods, while addressing this problem, does not change the main results of the analysis.

This appendix also further generalizes the model by relaxing the assumption that entrepreneurial talent is uniformly distributed and allows $\phi \geq 0$ to be distributed according to any cumulative distribution function $F(\phi)$.

C.1 The closed economy equilibrium

Assume the existence of a perfectly competitive sector where perfectly homogeneous tradable goods are produced. Labor is the only input and the production function in the tradable sector is $q = \alpha L$. The tradable good enters symmetrically in the Dixit-Stiglitz preferences of the agents so that its demand is $q = Y p_q^{-\theta}$, where the *numeraire* is, as in the main text, the price index of non-tradable goods.

In equilibrium $p_q = \frac{w}{\alpha}$ must hold. Being the supply of good q infinitely elastic at $p_q = \frac{w}{\alpha}$, the quantity produced in the tradable sector is determined by consumers' product demand. Therefore, the demand for labor in the tradable sector is given by $L_q = \frac{q}{\alpha} = \frac{Y}{\alpha} p_q^{-\theta} = \alpha^{\theta-1} Y w^{-\theta}$. Assuming, as in the main text, $\theta = 2$ we have that:

$$L_q(z) = \alpha \frac{Y}{w^2} = \alpha z \tag{19}$$

We can now derive the aggregate labor demand equal to the sum of the demand for labor in the tradable and non tradable sectors. From (7) and (19) we have that:

$$L_{D}(z) = L_{q}(z) + F(z-1) \times z + \int_{0}^{z-1} \phi dF(\phi)$$

= $\alpha z + F(z-1) \times z + \int_{0}^{z-1} \phi dF(\phi)$ (20)

From (20) the labor demand may be rewritten as follows:

$$L_D(z) = \begin{cases} \alpha z & \text{If } z \le 1\\ \alpha z + F(z-1) \times z + \int_0^{z-1} \phi dF(\phi) & \text{If } 1 \le z \end{cases}$$
(21)

Thus, differently from the case analyzed in the text, for $z \leq 1$ the labor demand is not zero as some labor is hired in order to produce tradable goods.³⁵ Of course, in any equilibrium where both tradables and non tradables are produced, it must be that z > 1. Thus, being the labor supply $L_S(z) = 1 - F(z-1)$ the labor market equilibrium condition that pins down the value of z is:

$$1 - F(z - 1) = \alpha z + F(z - 1) \times z + \int_{0}^{z - 1} \phi dF(\phi)$$

$$1 = \alpha z + F(z - 1) \times (z + 1) + \int_{0}^{z - 1} \phi dF(\phi)$$
(22)

Notice that the right hand side of (22) is continuously increasing in z and is equal to α if z = 1. Thus, provided that $\alpha < 1$, both tradables and non tradables are produced and the above equation uniquely identifies the autarchic equilibrium value of z denoted by Z_A .³⁶

The goods market equilibrium condition, that allows to separately identify the equilibrium values of the wage rate w and of the income Y, reads as follows.

$$Y = \alpha^2 z + 2F(z-1)zw \tag{23}$$

Given that the value of z is pinned down by the labor market equilibrium condition, the above equation allows to solve for w:

$$1 = \frac{\alpha^2}{w^2} + \frac{2F(Z_A - 1)}{w}$$
(24)

and therefore:

$$w_A = F(Z_A - 1) + \sqrt{[F(Z_A - 1)]^2 + \alpha^2}$$
(25)

³⁵Notice that we now allow for any $\phi \ge 0$, thus ϕ is not constrained to be smaller than 1 as in the main text.

³⁶Under the assumption of uniformly distributed talents one gets $Z_A = \frac{1-\alpha+\sqrt{(1-\alpha)^2+9}}{3}$

C.2 The open economy equilibrium

We solve for a symmetric steady state. When the economy is open the labor demand is given by the sum of the demand for labor of domestic and foreign entrepreneurs. The labor demand of domestic entrepreneurs is as in equation (21). The labor demand of foreign entrepreneurs is, as in the main text, given by:

$$L_D^f(z) = \int_0^{(z-\gamma-C)} (z+\gamma+\phi) dF(\phi)$$

= $F(z-\gamma-C) \times (z+\gamma) + \int_0^{(z-\gamma-C)} \phi dF(\phi)$ (26)

As in the main text, an equilibrium with FDI requires $1 \leq \gamma + C \leq z$. In this case (which holds true if $\gamma + C$ is low enough, namely $\gamma + C \leq Z_A$), the labor market clearing condition is:

$$1 - F(z - 1) = \alpha z + F(z - 1) \times z + \int_{0}^{z - 1} \phi dF(\phi) + F(z - \gamma - C) \times (z + \gamma) + \int_{0}^{(z - \gamma - C)} \phi dF(\phi)$$

$$1 = \alpha z + F(z - 1) \times (z + 1) + \int_{0}^{z - 1} \phi dF(\phi) + F(z - \gamma - C) \times (z + \gamma) + \int_{0}^{(z - \gamma - C)} \phi dF(\phi)$$
(27)

Also in the open economy case the right hand side of (27) is continuously increasing in z and is equal to α if z = 1. Thus, again, provided that $\alpha \leq 1$ the above equation uniquely identifies the equilibrium value of z denoted by $Z^*(C)$. Notice that the labor demand can never be smaller than in autarchy. As the supply of labor is not affected by the possibility of cross-border investments, in any equilibrium with multinational entrepreneurs (i.e., whenever $\gamma + C < Z_A$), it follows that labor is now relatively more expensive, i.e. $Z_A \geq Z^*(C)$.

Again, the goods market equilibrium condition allows to disentangle Y and w. The equilibrium condition reads as follows:

$$Y = \alpha^{2} z + 2zw \left[F(z-1) + F(z-\gamma - C) \right]$$
(28)

and therefore:

$$w^{*}(C) = F\left(Z^{*}(C) - 1\right) + F\left(Z^{*}(C) - \gamma - C\right) + \sqrt{\left[F\left(Z^{*}(C) - 1\right) + F\left(Z^{*}(C) - \gamma - C\right)\right]^{2} + \alpha^{2}}$$
(29)

A steady state with a lower C (more similar entrepreneurial environments) displays higher levels of w and Y.

The first result follows directly from equation (29 from which it is apparent that the wage rate is increasing in the fraction of (domestic and foreign) entrepreneurs, which in turn is decreasing in $C.^{37}$

The second statement must be true since both productivity (as a consequence of low talented entrepreneurs exiting the market) and the fraction of workers goes up as C decreases. Thus, output must go up.

Finally, we cannot claim in full generality that in steady state a lower C implies lower gross profits $\frac{Y}{w}$. This is due to the fact that we abandon the assumption of linearity in the distribution of talents and allow for a general CDF $F(\cdot)$. This can be seen rewriting equation (28) as follows:

$$\frac{Y}{w} = z \left[\frac{\alpha^2}{w} + 2 \left(F \left(z - 1 \right) + F \left(z - \gamma - C \right) \right) \right]$$
(30)

Recall that, as C goes down, z decreases while both w and the total number of entrepreneurs go up. Thus, two offsetting forces are at work on the right hand side of (30). In the model solved in the main text, with a linear $F(\cdot)$ and $\alpha = 0$, we are able to prove analytically that the negative effect prevails and $\frac{Y}{w}$ decreases as C goes down. While the inclusion of the tradable sector reinforces the negative effect, the presence of an arbitrary CDF does not allow to get analytical results. However, numerical results (available upon request) using different distribution functions (exponential, gamma, half-logistic and others) show that $\frac{Y}{w}$ does go down when C goes down.

³⁷Recall that a decrease in C raises the aggregate (i.e. the sum of domestic and foreign) labor demand, while triggering a decrease in the number of domestic entrepreneurs and an increase in the number of foreign entrepreneurs. For the overall labor demand to go up, it must be that the number of (very efficient) foreign entrants is larger than the number of (inefficient) domestic entrepreneurs leaving the market, because more efficient entrepreneurs demand a lower amount of labor than inefficient ones. Another way of seeing that a lower C raises the mass of entrepreneurs is that the number of varieties increases.